

Received 2 July 2024, accepted 5 August 2024, date of publication 12 August 2024, date of current version 21 August 2024. Digital Object Identifier 10.1109/ACCESS.2024.3442672

RESEARCH ARTICLE

Simultaneous Consideration of System Design and Post-Production Support Network **Decisions in the Context of a** Performance-Based Contract

ARACELI ZAVALA^{@1}, DAVID NOWICKI^{®2}, AND JOSE EMMANUEL RAMIREZ-MARQUEZ^{®3}

¹School of Engineering and Sciences, Tecnológico de Monterrey, Campus Guadalajara, Zapopan 45138, Mexico ²Department of Logistics and Operations Management, University of North Texas, Denton, TX 76203, USA ³Stevens Institute of Technology, Hoboken, NJ 07030, USA

Corresponding author: Araceli Zavala (araceli.zavala@tec.mx)

ABSTRACT In times of shrinking margins, researchers are looking for more cost-effective and profitable ways to improve the post-production support of large-scale, complex systems. There is an inherent tradeoff between the system design and the design's long-term support. A system's design largely determines its reliability that, in turn, influences the demands on its post-production support network necessary to maintain the proper use of the system over its intended, economic useful-life. Performance-based contracts are a successful financial instrument between suppliers and buyers for long-term support contracts. This research leverages the tenets of performance-based contracting, especially its foundation in transactional cost economics and management control theory, and agency theory, to develop and test an analytical model. This research proposes a novel, analytical model that maximizes the profit margin of a large-scale, complex system simultaneously considering its design and post-production support network. To date, these decisions are largely understudied. This model uses redundancy allocation to represent a design decision, and the post-production support network decisions are the location, quantity of spares, and logistics footprint. The post-production support network is a non-arboreal, multi-echelon sustainment network, and the design is a series-parallel configuration. The model is constrained by customer-specified, minimum reliability, mean-time-between-failure (MTBF), and a maximum logistics footprint (LF) measured in pounds. A metaheuristic algorithm was used to address the nonlinearity of the objective function and constraints. Afterward, a numerical example was solved, and comparative experiments were conducted to test the algorithm. The results showed a profit for the supplier of \$14,505.12 with 78.45 hrs of MTBF out of the 75 hours minimum allowed and a logistic footprint of 6,285.25 lb. out of the 10,000 lb allowed. The solution demonstrates the economic importance of system engineers, contract personnel, and program managers in understanding the inherent tradeoff space connecting the design and support of a system.

INDEX TERMS Inventory optimization, performance-based contracting, reliability, supply chain networks, systems theory.

I. INTRODUCTION

Increasingly the design community, often spearheaded by systems engineers, and the sustainment community is collaboratively looking for more cost-effective and profitable

The associate editor coordinating the review of this manuscript and approving it for publication was Xiao-Sheng Si $^{\square}$.

ways to simultaneously provide a better-performing system and improved post-production support to their customers. In times of shrinking margins, reduced funding, and increased competition, it makes sense that managers would seek innovative strategies to facilitate such competitive challenges. One such emerging strategy is performance-based contracting (PBC); especially in capital-intensive systems such as

© 2024 The Authors. This work is licensed under a Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 License.

For more information, see https://creativecommons.org/licenses/by-nc-nd/4.0/

aerospace, defense, energy, healthcare, highway/railway, etc [1]. PBC is otherwise known as "Performancebased logistics" (PBL) [2], "Availability Contracting" [3], "Contract for Availability" [4], "Performance-based Service Acquisition" [5], and "Outcome-based Contracting (OBC)" [6].

Traditionally, post-production contracts are consistent with Transactional Cost Economics (TCE), where suppliers perform maintenance on a system and the buyer provides remuneration for each maintenance action. With PBC, the underlying theoretical lens of TCE is now replaced through agency theory and the theory of incentives, where suppliers are financially incentivized to provide outcomes to the buyer. So, the supplier is no longer compensated for each maintenance transaction completed, but rather, rewarded or penalized based on system performance outcomes [7].

As a buyer-supplier contractual strategy, the use of PBC is present in public and private sectors [8], [9]. For example, PBC is used for the procurement and maintenance of highway and railway infrastructure [10], [11], [12], [13], [14], health and social care [15], energy [16], [17], manufacturing [8], defense [18], and aerospace [19]. The U.S. Department of Defense (DoD) has been using PBC as the preferred sustainment strategy since 2001 [20]. Using PBC across different industry sectors shows a 25-40% increase in reliability, a 10-20% reduction in the cost per unit of performance, and operational availability improvements [2], [21].

PBC has been successful in generating buyer value and the supplier-buyer network profitability; most notably for complex, hight-cost systems where the cost of sustaining the system over its economic, useful-life far exceeds its cost of research, development, and production [8], [22]. PBC aims to make a supplier deliver performance outcomes as specified by key performance indicators (KPI) for the contracted work while creating incentive schemes through longer contracts and opportunities for cost avoidance [22]. For example, Rolls-Royce created a "power-by-hour" (PBH) program for its airplane engines where it is responsible for the availability of the engines, performing the necessary engine maintenance, repair, and overhaul, and the buyers of the engines pay Rolls-Royce based on each flight hour it uses its engines [6]. Another company that can exhibit how PBC generates value is General Electric (GE). They implemented a condition-based monitoring and maintenance service agreement for their pressure control equipment service business instead of having a calendar and event-based maintenance. This allowed GE to share some of the operating risks with its customers and offered a risk-adjusted value proposition while gaining more control over the condition and maintenance of its equipment.

Redundancy in spare allocation for maintenance purposes is essential to secure high system reliability and operational availability. Solving spares allocation in a multi-echelon supply chain is a complex problem because every decision made in one echelon affects both upstream suppliers and downstream buyers within the supply chain. To date, spares allocation in a multi-echelon supply chain has been solved to minimize logistics costs or maximize the availability of spares through efficient inventory placement between echelons [23]. Researchers have used a variety of constraints or requirements to solve the spares allocation problem, such as lead time variability [24], a number of service engineers needed to do the replacement [25], disruptions in different echelons of the supply chain [26], budget constraint [27], or demand variability [28], among others.

One of the main objectives of PBC is to reduce the Mean Time Between Failure (MTBF) and, consequently, improve the service process and reduce costs [7]. However, studies have focused on balancing reliability and maintenance effectiveness. A broader perspective considering sustainability aspects in multi-year contracts or including resilience is missing [29]. In addition, environmental sustainability aspects must emerge as a requirement in PBC. Wang et al. [30] developed a model for deteriorating production systems operating under PBC. In this model, the customer offers PBC payments while the supplier implements Conditioned-based maintenance (CBM), which represents periodic inspections and sets an investment for spare parts. They proposed a two-staged heuristic approach to maximize utilities for both stakeholders. PBC strategy leverages long-term relationships to turn year-after-year sustainment cost avoidance into a potential return on investment [1]. To that end, this manuscript's objective is to propose a new algorithm that, when solved, allows system engineers and managers to understand the impact that design and, mainly, system redundancy have on total life-cycle costs associated with spare inventories. The algorithm, coupled with the correct support strategy, broadens the effective "market potential" through outcome-based multi-year contracts, such as PBC.

Systems engineers are increasingly adopting a System-of-Systems approach when describing the interaction between the design of a system, the system operation, and the post-production support network necessary to sustain the operation of that system. The system and its necessary post-production support network combined provide outcome-based value (e.g., operational availability) to the buyer of the system at a profit that is acceptable to the supplier. There is substantial research in the extant literature when system design (redundancy allocation) and post-production support network (quantity and location of spares) decisions are considered separately. However, to date, the complexities of simultaneous consideration of system design and post-production support network decisions are largely understudied. A critical aspect of system design involves determining the impact of system design and redundancy on sparing. These decisions are critical to providing customer value and maximizing profit through multi-year, fixed-price contract cost avoidance [1].

This paper addresses these complexities by defining a novel, analytical model that maximizes the profit margin of a large-scale, complex system simultaneously considering its design and post-production support network. This model uses redundancy allocation to represent a design decision, and the post-production support network decisions are the location, quantity of spares, and logistics footprint (LF). The post-production support network is a non-arboreal, multi-echelon sustainment network, and the design is a series-parallel configuration. Specifically, the model is constrained by buyer-specified, minimum reliability, measured as the mean-time-between-failure (MTBF); and a maximum logistics footprint (LF), measured in pounds.

As aforementioned, redundancy allocation models in the extant literature focus only on component reliability and not the effects reliability (or non-reliability) has on spares [31]. As components fail, demands are generated on a system's post-production support network that includes spares. Our proposed analytical model considers this linkage.

The primary research question guiding this study is:

• Is it more advantageous to develop system support mechanisms and system design in conjunction rather than independently?

Along with the following sub-questions:

- 1) How does the integrated development of postproduction support network and design impact overall system performance and reliability?
- 2) What are the cost implications of a conjoint development approach compared to separate development?

Compared with the extant literature, the main contributions of the present paper are summarized as follows:

- We develop an analytical model that maximizes the profit margin of a large-scale, complex system while considering its design and post-production support network.
- We propose a meta-heuristic algorithm that determines the level of redundancy in each subsystem and the quantity and location of spares to support the system's operation.
- We include environmental sustainability aspects regarding the logistics footprint as a constraint in the model.
- We evaluate the proposed model through a numerical example and compare different feasible solutions to provide a better understanding of the model and its managerial insights.

The remainder of this paper is organized as follows: Section II reviews the relevant redundancy allocation problem and PBC literature. Section III presents the notation used in this paper. It also describes the relationship between the system design and its post-production support network and briefly describes the performance metrics used in the analytical model. Section IV explains the analytical model and the proposed meta-heuristic algorithm to solve it. Section V shows a numerical study implementing the model and its results. In addition, we present several managerial insights in Section VI. Finally, this paper concludes in section VII with conclusions and future research.

II. BACKGROUND

The motivation of the present research and the relevant existing literature are presented in this section.

A. REDUNDANCY ALLOCATION PROBLEMS

Redundancy allocation is a system design problem that determines whether redundancy is necessary and, if necessary, where to locate it in the design. The intent is to optimally assign redundancy within the system design to meet a buyer-specified reliability target at a minimum cost or to maximize the system reliability given a budget constraint. The reliability allocation problem (RAP) is best addressed early in the design process, with the greatest opportunity to influence cost and performance.

Kennedy et al. [32] did a literature review on spare parts inventory for supporting infrastructures. They pointed out that a critical design decision involves choosing the ideal number of redundant equipment to design into the primary system. If there are sufficient redundant parts to achieve a reliability design target, it may be possible to replace all failed parts at a single point in time. There are other literature reviews on spare parts inventory, mostly focused on forecasting for inventory planning and spare parts classification [23], [33]. A recent literature review on solving the RAP reviewed 275 papers [34] where authors used different system configurations, performance measures, problem formulations, and solution methods. For example, several authors used a genetic algorithm (GA) to achieve a solution or a modified version of GA [35], [36], and others developed a solution space reduction procedure and a branch-and-bound algorithm to obtain a solution, which implicitly enumerates the solutions [37]. The latter method has yielded optimal results but requires potentially inefficient enumerations. Other approaches to solve the RAP include several types of simulations, from Monte Carlo simulation to stochastic simulation combined with neural networks and GA [38], [39], [40]. A recent review paper show the spares allocation gap between modelling and application and, between theory and practice. In their search, they included maintenance-policy classes and inventory optimality rules because they argue that the gap between theory and practice is growing due to the increasingly complicated mathematical problems [41].

Hemmati et al. [42] solved the RAP as a multi-objective problem, maximizing the MTBF and minimizing the system's total cost. Their multi-objective algorithm, named Multi-Objective Harmony Search (MOHS), saves CPU time compared to the Non-dominated Sorting Genetic Algorithm (NSGA-II) and Non-dominated Ranked Genetic Algorithm (NRGA). In this study, we implemented Monte Carlo simulation to generate potential series-parallel system design configurations. Later, the algorithm evaluates whether this design satisfies the MTBF and LF constraints. Allocation of spares in isolation, without regard to the total ownership cost, invariably leads to a suboptimal design. Hadel et al. [43] point out that all system-level requirements must be addressed simultaneously to ensure those requirements are optimally addressed within design constraints. They developed a multi-step technique. First, a Customer Quality Model (CQM) is created that uses a modified Quality Function Deployment. Using the CQM, a numerical allocation of customer-specified, system-level requirements is performed. With PBC, the focus is now on the supplier to reduce the system design's present and future economic considerations while guaranteeing buyer-specified system performance.

B. PERFORMANCE-BASED CONTRACTING

PBC focuses on the supplier's post-production support of the system the supplier delivered to the buyer [44]. PBC is especially relevant for sustaining large-scale complex systems present in the aerospace, defense, aircraft, and utility industry sectors. This strategy has been adopted due to an increase in the number of end-users, operators, original equipment manufacturers (OEMs), and sustainment service providers who are faced with longer service life cycles, higher costs, and demand for improved system performance [45]. Practitioners attribute billions of dollars in post-production support savings to PBC [46].

Post-production support costs are a significant and profitable part of the economy [22]. Using a PBC strategy, supplier networks, and their buyers are adopting the traditional return on sales of repair, spare, and overhaul revenue model. Instead, these buyer-supplier partnerships engage in long-term contracts that tie compensation to performance using a return on investment business strategy. This makes sense, especially for those high-cost, repairable, complex systems (e.g., high-speed rail, defense systems, telecommunication) whose support costs often exceed their research, development, and production costs [22]. Savvy system operators and buyers see the line-by-line and yearby-year sustainment costs associated with these systems as a tool to encourage the supplier network to invest in reducing system cost while improving system performance for the buyer [22]. In the traditional post-production support strategy, not only the supplier is charged at the beginning of the project but also charged for each service it performs to sustain the system, in which the profit is linked to the post-production services of the sale of parts, equipment, transportation, and labor [21], [47].

Throughout the 1990s, companies like Toyota and General Electric demonstrated how reliability and quality addressed not only fundamental warranty issues but also increased revenue, decreased costs, and built brands [48]. PBC links design and supply chain strategy with the buyers' total life-cycle cost and performance experience. By linking supplier network decisions to life-cycle operational costs, PBC is a mechanism to focus engineers, operators, and supply managers on identifying cost-saving opportunities.

Understanding the year-to-year costs associated with inventory, transportation, warehousing, repair, purchasing, and more, and linking those costs to system design and redesign through a return on investment incentive structure, provides the supplier network with a significant potential market [1], [45].

Strategies such as PBC adopt detailed year-to-year profit maximization through a cost avoidance perspective when creating design decision trade space. This view has caused designers, suppliers, OEMs, operators, and buyers to rethink their reliability, redundancy, and spare decisions during the system's design, production, operation, and support. The emergence of PBC and its multi-year performance-based strategies has highlighted research and practice gaps. Academicians and practitioners have noted that these strategies significantly increase the complexity of design decisions [1], [22], [49], [50], [51].

Academic research into PBC strategies continues to emerge and holds promise [1], [22], [49], [52], [53], [54]. To date, PBC research has focused on the business aspects of PBC and, to a lesser extent, mathematical models in support of a holistic approach to PBC. For example, Sols et al. [49] developed an n-dimensional, PBC model. Kim et al. [22] and Nowicki et al. [55] focused their research on the post-production support of a system design where inventory allocation problems were studied under a PBC.

PBC has driven a requirement for new algorithms from which to guide system design decisions that tradeoff sparing, repairing, network structure, and network cost. Previously, a supplier's post-production approach to sustainment was uncoordinated with many groups within a supplier's organization, each providing parts and services to the buyer procuring the system. Buyers pay suppliers for each maintenance, repair, overhaul, and spare support transaction, which represents a shift in how buyers purchase post-production support [56]. In contrast, a PBC is a post-production strategy that provides buyer-supplier incentives to optimize system performance through long-term contractual arrangements with clear lines of authority and responsibilities. The supplier is paid for delivering an outcome to the buyer (e.g., \$/flight hour) rather than being paid to provide spares and repairs (e.g., per maintenance actions) [45], [57].

Under a PBC approach, the system integrator, which typically is the OEM, is responsible for providing an outcome for a fixed fee and incurs the cost associated with that outcome. In other words, the OEM is now responsible for the total supply chain cost required to meet a certain buyer-specified service level. As a result, the revenue model associated with production shifts significantly in comparison to traditional sustainment [21]. In traditional sustainment, the OEM and supplier increase their profits when they sell more spares or repairs, a "return on sales" revenue model [57]. On the other hand, in PBC, the integrator makes the greatest profit when the system does not fail or when the cost of maintaining the system is minimized. This drives a return on investment business model. Profits are maximized

when decisions and designs are made to minimize the total life-cycle cost over the PBC contract period.

PBC's return on investment model creates a need for more accurate sparing, inventory, reliability, and redundancy algorithms. The supplier has a much larger potential return on investment concerning total life-cycle cost. While in a warranty approach, the supplier is typically responsible for just the cost of the spare part, in a PBC approach, the supplier is responsible for the total inventory, warehouse, transportation, maintenance, training, and support equipment required to recover that system when it fails. Thus, PBC design and profit decisions entail complicated multi-objective solutions to tradeoff sparing, maintenance level, and redundancy design. The industry is searching for ways to more effectively invest dollars in reducing total system life-cycle costs [9]. Under a PBC, such improvement in investment decisions represents not only increased profit but also lower out-year resource consumption, which means greater sustainability. As a consequence, suppliers will achieve an increase in their return on investment as the length of the PBC increases, leading to improved forecast accuracy. Accurate historical data helps suppliers efficiently manage performance risks [58]. Glas et al. [59] studied 12 risks associated with PBC, from dependency, misperformance, and misbehavior relationships to a lack of qualified personnel. They conclude that the higher a firm's experience with PBC, the lower its perception of risk. In this study, they consider content risk, business relationship risk, and operational risk. Researchers attempted to evaluate the impact of performance measurement in PBC. There is no exact way to evaluate performance within PBC because it depends on the industry sector and the contractual agreements [60]. Another question arises relating to the payments and the supplier's net profit. A study by Patra et al. [61] applies a first-order autoregressive moving average (ARMA) process to analyze a supplier's future net profit for linear and non-linear revenue functions. Ultimately, this approach determines the optimal system availability a supplier can provide to a buyer when the probability of loss is considered. Outsourcing has been used to minimize costs in complex projects; however, it is affected by organizational arrangements and the goal orientation of the projects [62]. We use the penalty, dead, and reward zones between the operational availability of the system and the amount of money the buyer pays the supplier [55].

III. PROBLEM DESCRIPTION, DECISION VARIABLES, PARAMETERS, ASSUMPTIONS, AND SYSTEM PERFORMANCE METRICS

This section explains the problem description, notation, assumptions, and system performance metrics.

A. PROBLEM DESCRIPTION

This research proposes a novel, analytical model that maximizes the profit margin of a large-scale, complex system simultaneously considering its design and post-production support network. This model uses redundancy allocation to



FIGURE 1. System Design and its Post-production Support Network.

represent a design decision, and the post-production support network decisions are the location and quantity of spares and their logistics footprint. The post-production support network is a non-arboreal, multi-echelon sustainment network, and the design is a series-parallel configuration, refer to Fig. 1. The model is constrained by buyer-specified, minimum reliability, measured as the MTBF, and maximum LF, measured in pounds.

The system design is a series-parallel configuration (\mathbf{x}) where the system's subsystems (i^{th} subsystem of the system, where i = 1, 2, ..., |I| are in series and each subsystem consists of components $(j^{th}$ component of the i^{th} subsystem, where $j_i = 1, 2, ..., |J_i|$ in parallel. The parallel components are actively redundant. These systems are often present in the consumer electronics industry where systems use standard components, redundantly configured to improve system reliability [63]. The system presents a complex redundancy allocation problem that is combinatorial difficult. Coit and Liu [64] derive a component selection algorithm when commercial, off-the-shelf components are considered that have a known cost, reliability, and weight. Also, they argue that redundancy allocation during system design is difficult, largely due to its large combinatorial space. The system's post-production support network (s) is a non-arboreal, multiechelon sustainment network that does not consider lateral resupply. The post-production support network is represented by a set of locations L_e that reside within a set of supply echelons (E); more specifically defined as the l^{th} location within the e^{th} echelon, where $l_e = 1, 2, \ldots, |L_e|$ and the e^{th} echelon in the post-production support network, where $e = 1, 2, \ldots, |E|.$

The system operates until one of its subsystems fails. Then, a corrective maintenance action is initiated, in which all components within the failed subsystems are removed and replaced with functioning spare components. This is essentially a one-for-one replenishment policy valid for largescale, repairable systems with expensive subsystems that fail infrequently [65], [66]. If available, spare components stored at the operational location of the system are used $(l_{|E|})$. Under this scenario, the average time to operationally restore the system is the average time to remove and replace all components in the j^{th} subsystem that failed $\mu_{l_{|E|}} = \sum_{j=1}^{|J_i|} \mu_{jil_{|E|}}$. If there are no spares in stock at this location, then a replenishment action is initiated to move inventory from another supply location and echelon of support. If replenishment is necessary, the total delay includes the replenishment lead time, $LT_{j_{l_e}}^R$, in addition to the components' removal and replacement time, $\mu_{l_{|E|}}$. $LT_{i_{j_e}}^P$, the newly procured spare enters the support pipeline at the aggregate or central echelon of maintenance (e = 1).

With a series-parallel system configuration, the subassembly failure rate is not the demand rate on the post-production support network. A subassembly fails when all redundant components fail; therefore, a sub-assembly failure triggers multiple remove-and-replacement maintenance actions, one for each of its components. As a consequence, the component demand rate is the subassembly failure rate multiplied by the number of redundant components in the subassembly, $d_{j_il_e}(t) = x_{j_i} \times \lambda_{j_il_e}$. To determine the subassembly failure rate, a random variable *T* is defined to denote the time to failure and it is assumed that it follows a Poisson process, where the expected value of *T*, or expected time to failure, is $E[T] = MTTF_{jl_e}$ and $\lambda_{jl_e} = 1/MTTF_{jl_e}$. With the additional assumption of perfect renewals, the $E[T] = MTBF_{jl_e}$ and $\lambda_{jl_e} = 1/MTBF_{jl_e}$.

Upon failure, delays negatively impact the system's performance. The duration of the delay depends on the location and quantity of spares in the system's post-production support network. At a minimum, the system will experience a delay in removing and replacing the failed components of the failed subsystem. If spares are present in the support network, there is an additional replenishment delay. The maximum delay occurs when a procurement action is necessary to repopulate the spare pipeline.

If a component demand is unfulfilled, a backorder occurs. An unfulfilled component demand occurs when the number of demands exceeds the amount of inventory on hand at the location where the subassembly failed. The size of the backorder is the difference between the component demand and its available inventory, $d_{j_i l_e}(t) - s_{j_i l_e}$. Backorders negatively impact operational availability, a system performance metric that is agreed upon between a supplier and buyer in a PBC. The backorder is a random variable, and its expected values are defined as:

$$BO(d_{j_i l_e}(t)|s_{j_i l_e}) = \begin{cases} d_{j_i l_e}(t) - s_{j_i l_e} & \text{if } d_{j_i l_e}(t) \ge s_{j_i l_e} \\ 0 & \text{otherwise} \end{cases}$$

This paper uses Sherbrooke [67] expected backorder to operational availability transformation, defined as:

$$A_0 = 100 \prod_{l_E}^{|E|} \prod_{i=1}^{|I|} \left\{ \frac{(1 - E[BO(d_{j_i l_e}(t)|s_{j_i l_e})])}{n_{L_{|E|}}} \right\}^{n_{L_{|E|}}}$$
(3)

Sherbrooke [67] transforms the backorders of the components at the system operation locations. The systems operate and fail in locations $l_{|E|} = 1, 2, ..., |L_{|E|}|$. The number of operating systems $(n_{l_{|E|}})$ are input into the model. Operational availability is a critical system performance metric that is commonly specified in a PBC. This study uses operational availability, reliability, and LF in the proposed meta-heuristic analytical model. These three system performance metrics are discussed in the next section.

B. DECISION VARIABLES AND PARAMETERS

The notations of the model are defined as follows: **Indices**

- *i* i^{th} subsystem of the system configuration, where i = 1, 2, ..., |l|.
- j_i j^{th} component of the i^{th} subsystem, where $j_i = 1, 2, ..., |J_i|$.
- *e* e^{th} echelon in the post-production support network, where e = 1, 2, ..., |E|, and
- l_e l^{th} location within the e^{th} echelon, where $l_e = 1, 2, \dots, |L_e|$.

Sets

- *I* Set of subsystems in the system configuration.
- J_i Set of components in the i^{th} subsystem.
- *E* Set of echelons in the post-production support network, and.
- L_e Set of locations in the e^{th} echelon in the post-production support network.

Variables

- $MTBF^T$ buyer-specified mean time between failure for the system.
- *LF^T* buyer-specified logistics footprint target for the supplier-provided post-production support network.
- $n_{l|E|}$ number of systems operating at location l_e in the echelon |E|.
- $c_{i_j l_e}$ cost of the j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon of the post-production support network.
- $LF_{i_j l_e}$ logistics footprint of the j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon of the post-production support network.

- $\lambda_{i_j l_e}$ failure rate of the j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon of the post-production support network.
- μ_{ijle} repair rate of the j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon of the post-production support network.
- $r_{i_j l_e}$ probability of repairing *j*th component in the *i*th subsystem at the *l*th location in the *e*th echelon of the post-production support network.
- $LT_{i_j l_e}^R$ replenishment lead time of the j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon, and.
- $LT_{i_{j}l_{e}}^{P}$ procurement lead time of the *j*th component in the *i*th subsystem located in the *l*th location in the *e*th echelon.

C. ASSUMPTIONS

The research problem assumptions are as follows:

- 1) The post-production support network is a non-arboreal, multi-echelon sustainment network without lateral supply between locations. (detailed information can be found in [68].
- 2) The system design is a series-parallel configuration.
- 3) There is a one-for-one replenishment policy.
- 4) To calculate operational availability, we use Sherbrooke [67] expected backorder, as shown in Eq. 3.
- 5) The subassembly failure rate follows a Poisson process with perfect renewals after replacing a component.

Parameters

Х	Vector that represents the system configu-
	ration.
s	Vector that represents the post-production
	support network.
r.	The number of the i_{i} component in the i_{i}

 x_{ij} The number of the j_{th} component in the i_{th} subsystem.

 $s_{i_j l_e}$ Stock level of the j_{th} component in the i_{th} subsystem located in the l_{th} location in the e_{th} echelon of the post-production support network.

- MTBF(x) Mean time between failure of the system configuration **x**.
- LF(x, s) Logistics footprint of the spares **s** needed to support system configuration **x**.
- $\Pi(x, s)$ Supplier profit when the buyer purchases system **x** and the supplier sustains system **x** with post-production support network **s**.
- $R[A_0(x, s)]$ Supplier revenue when the buyer purchases system **x** and pays for system availability $A_0(x, s)$.
- $A_0(x, s)$ Operational available of system **x** and its post-production support network **s**.
- C(x, s) Supplier cost to design and produce system **x** and its corresponding post-production support network **s**.

- V(x, s) Buyer value of system **x** and its post-production support network **s**.
- P(x, s) Aggregate penalty derived if system **x** and its post-production support network **s** do not meet the buyer-specified logistics footprint and MTBF targets.
- $P^{MTBF}(x, s)$ Penalty derived if system **x** and its post-production support network **s** do not meet the buyer-specified MTBF target.
- $P^{LF}(x, s)$ Penalty derived if system **x** and its post-production support network **s** do not meet the buyer-specified logistics footprint target.
- $BO(s_{ijl_e})$ Random variable that presents the backorders of the j_{th} component in the i_{th} subsystem located in the l_{th} location in the e_{th} echelon of the post-production support network with stock level s_{ijl_e} .
- $E[BO(s_{i_j l_e})]$ Expected number of backorders of the j_{th} component in the i_{th} subsystem located in the l_{th} location in the e_{th} echelon of the post-production support network with stock level $s_{i_j l_e}$.
- $d_{i_j l_e}(t)$ Expected number of demands of the j_{th} component in the i_{th} subsystem located in the l_{th} location in the e_{th} echelon of the post-production support network during time interval t.

D. SYSTEM PERFORMANCE METRICS

Traditionally, system design performance focuses on engineering capabilities, often excluding its inherent reliability, maintainability, and supportability (RMS). For example, aircraft design generally focuses on takeoff and landing, turns, straight and level-flight in cruise, and climb. A more detailed example is an aircraft structure's stiffness and strength-to-weight ratio. The RMS of a system is central to risk and quality of its design, both critical aspects of performance. Kratz et al. [69] argue that engineering performance and RMS must be simultaneously considered and introduce the concept of System Design and Operational Effectiveness (SDOE). SDOE, as shown in Fig. 2, illustrates a holistic approach to system design where engineering performance, RMS, and process efficiency are considered, not only in the design phase of a system but throughout its entire life cycle (design, production, sustainment, and endof-life).

From an economics perspective, SDOE uses system lifecycle cost, or total cost of ownership (TCO). A system design's TCO considers the development and design costs and the costs incurred when the design is produced, operated, and maintained. The RMS of a system design determines its frequency of failure (reliability, often measured as mean time between failures), the duration of time it takes to return the system to operation (maintainability, often measured as mean



FIGURE 2. SDOE holistic perspective system design, adapted from [69].



FIGURE 3. Committed vs. actual expenditures, adapted from [71].

time to perform a maintenance task), and the amount of time waiting for the necessary resource to return the system to operation (supportability, often measured as mean logistics delay time). Designing for RMS, their complex interactions, and their impact on downstream costs and revenues form the basis for TCO and inform a firm's procurement and asset management strategy [70].

Systems thinking is core to systems engineering principles and embraces a holistic approach that considers RMS, engineering performance, process efficiency, and TCO early in the design process [69]. Design decisions commit support resources and downstream costs. The earlier these linkages are understood, the better the design and post-production support network. Fig. 3 shows committed expenditures versus actual expenditures over a system lifecycle. Fabrycky [71] suggest that 85% of a system's TCO is committed early in the design process, with only 15% of the TCO's actual expenditures, consistent with Asiedu and Gu [70] assertion that minimizing TCO should focus on minimizing the system's operation and maintenance cost. When system engineers take a holistic perspective, understating the detailed linkage between system design to its enabling infrastructure, a healthier design emerges [72], [73], [74].

As shown in Fig. 4, operational availability A_0 is measured as the proportion of time a system is operating (system uptime) relative to the proportion of time it is operating or down (uptime + downtime).



FIGURE 4. System design impact on rms and availability, adapted from [69].

The system is operational until it is down, either due to failure or a preventative maintenance task. This paper assumes a system is down only due to failure. The system remains down until it is restored to full operational status. The time to restore the system is the wait time to get all the resources (e.g., labor, material, and support equipment) necessary to perform the repair and the repair time. The operational time until failure is determined by the system's reliability, represented as the time to failure. The wait time, as determined by the quantity and location of the post-production support network resources, depends on the system's supportability, represented as the time to support. The time to repair is determined by the system's maintainability, represented as the time to maintain.

The system's operational availability is a metric that includes the system design's reliability, maintainability, and supportability. It is defined as the ratio of uptime over the summation of uptime and downtime, refer to Fig. 4. The uptime is a function of the system design's reliability, and the downtime is a function of the system design's maintainability and supportability. The denominator of A_o is a function of the system uptime and downtime. The downtime is the expected delay in waiting on resources (facilities, equipment, spares, transportation, and labor) to perform the maintenance actions to fully restore system operations [75].

Component reliability is central to our model as it determines the demand for component spares, which in turn determines, in part, the system availability, all at a cost over the system lifecycle. We assume the time to failure follows a Poisson process, and perfect renewals occur after a component is replaced or repaired. The Poisson process is selected because the average time between failures is known; however, the exact timing of failures is random. In addition, the arrival of a failure is independent of the failure before. As such, we can now define a component's failure rate as $\lambda_{i_j l_e} = \frac{1}{MTTF_{i_j l_e}}$ where $MTBF_{j_i l_e}$ is the mean time between failures of j^{th} component in the i^{th} subsystem in the l^{th} location within the e^{th} echelon. Also, the component reliability is $R_{x_{i}}(t) = e^{-\lambda_{j_{i}l_{e}}t}$. The system design's configuration is defined by a vector $\mathbf{x} = (x_{1_1}, \ldots, x_{j_i}, \ldots, x_{|J_l|}),$ where $x_{i} \in (0, 1)$. The system design's configuration is series-parallel, where the components of a subsystem are in parallel (redundant) and the subsystems are in series. The reliability of the series-parallel system is $R_x(t) =$ $\prod_{i=1}^{|I|} [1 - \prod_{i=1}^{|J_i|} [1 - R(x_{j_i}]^{x_{j_i}}].$ Assuming the reliability for

all components within a given subsystem are the same $(n_j = \sum_{j_i}^{|J_i|} x_{j_i})$, the subsystem reliability is $R(\mathbf{x}) = [1 - (e^{-\lambda_{j_i le}t})^{n_j}]$. The subsystem reliability drives system availability, and the component failure rates create demands for the post-production support network resources. These demands, along with the post-production support network's LF, provide the performance metrics for our analytical model.

LF is the collection of resources (spares, materials, support equipment, and labor) that must be collocated with the system. As such, these resources also need to move (so, also need transportation resources) with the system when the system moves locations. LF is often measured in weight and volume, so the smaller, the better [75]. The focus on LF is prominent in the U.S. DoD, where there is a tactical advantage to having an agile, mobile system. Its support resources must be equally agile and mobile for an agile, mobile system. A reduced LF is socially responsible [76] and conforms to regulatory policies [77]. Having a higher quality of buyer-supplier relations helps to achieve a better outcome related to green practice [78]. For this research, we define LF as

$$LF(x,s) = \sum_{e}^{|E|} \sum_{l}^{|L_{e}|} \sum_{i=1}^{|I|} \sum_{j=1}^{|J_{i}|} LF_{j_{i}l_{e}} \times s_{j_{i}l_{e}}$$
(4)

There are costs associated with the system design and its supporting infrastructure (reliability, maintainability, supportability, availability, and LF). These costs occur throughout a system's economically useful life. We focus on the system's design, production, and support costs. The design and production costs are reflected in the components, subsystems, and system price. The support costs are the cost to provide spares (location and quantity), so the systems' post-production support network can provide the contractually agreed upon system performance from the supplier to the customer. The system design and post-production support network costs are:

$$C(x,s) = \sum_{e}^{|E|} \sum_{l=1}^{|L_e|} \sum_{i=1}^{|I|} \sum_{j=1}^{|J_i|} c_{j_i l_e} + (s_{j_i l_e} + n_j x_{j_i l_e}).$$
(5)

Systems engineers, design engineers, engineering managers, and program managers use operational availability, LF, and lifecycle costs to guide system design decisions. An inherent trade-off exists between reliability, maintainability, supportability, LF, and lifecycle cost. We introduce a meta-heuristic model that incorporates this tradeoff space so that more informed design decisions are made. This analytical model concurrently considers design and post-production decisions in the context of PBC.

E. THE MODEL

We develop a novel, analytical model that maximizes a largescale, complex system's profit margin while considering its design and post-production support network. This model uses redundancy allocation to represent a design decision,



FIGURE 5. Relationship between supplier revenue and system availability, adapted from [55].

S

and the post-production support network decisions are the location and quantity of spares and their logistics footprint. The post-production support network is a non-arboreal, multi-echelon sustainment network, and the design is a series-parallel configuration. The model is constrained by buyer-specified, minimum reliability, measured as the MTBF, and a maximum LF, measured in pounds. The mathematical formulation of the general model follows:

max
$$\Pi(x, s) = R[A_0(x, s)] - C(x, s)$$
 (6a)

a.t.
$$MTBF(x) \ge MTBF^{T}$$
 (6b)

$$LF(x,s) \le LF^T$$
 (6c)

The model determines the optimal system design configuration **x** and post-production support network resources **s** that maximize the supplier's profit without violating the buyer's reliability and logistics footprint constraints. Specifically, the model holistically examines the system design and its post-production support network to determine the level of component redundancy in each subsystem $\mathbf{x} = (x_{1_1}, \ldots, x_{j_i}, \ldots, x_{|J_l|})$, where $x_{j_i} \in (0, 1)$, and the quantity and location of spares to support the operation of the system $s = (s_{1_1 1_0}, \ldots, s_{j_l l_e}, \ldots, s_{|J_l||L_E|})$, where $s_{j_l l_e} \ge 0$.

In a PBC, the buyer compensates the supplier based on the system performance. Brown and Burke [79] introduce the notion of rewards and penalties tied to the system's performance. We use the penalty, dead, and reward zones paradigm to model the linkage between the operational availability of the system and the amount of money the buyer pays to the supplier (revenue) [55], as shown in Fig. 5. The same step revenue function was used to reflect the rewards and penalties on the suppliers' efforts to increase the availability [80].

The supplier's revenue $R[A_o(x, s)]$ is shown in equation 6a, and the supplier's cost was previously described in equation 5. The supplier revenue consists of four increments of payments the buyer pays for system availability ranging from a minimum acceptable availability to a maximum value, where the supplier will not pay any extra for exceeding this availability. We assume a linear relationship between system availability and supplier revenue within each increment. Equation 6b shows the minimum reliability measured as the mean-time-between-failure while equation 6c represents the maximum allowed logistics footprint, measured in pounds.

$$\begin{aligned}
& [A_0(x, s)] \\
&= \begin{cases}
0 & \text{if } 0 \le A_0 < A_{min} \\
\alpha_1 + \beta_1 \times (A_0 - A_{min}) & \text{if } A_{min} \le A_0 < A_{penalty} \\
\alpha_2 + \beta_2 \times (A_0 - A_{penalty}) & \text{if } A_{penalty} \le A_0 < A_{accept} \\
\alpha_3 + \beta_3 \times (A_0 - A_{accept}) & \text{if } A_{accept} \le A_0 < A_{max}
\end{aligned}$$
(7)

Unfortunately, solving the aforementioned model using traditional closed-form optimization techniques such as integer programming is unsolvable since the objective function and the constraints are non-linear. For this reason, the following section explains a meta-heuristic algorithm that solves the model.

IV. SOLUTION APPROACH

We develop a meta-heuristics algorithm to solve this model, that is, to optimally determine the system design configuration and post-production support network resources that maximize the supplier's profit without violating the buyer's reliability and LF constraints. Specifically, the model holistically examines the system design and post-production support network to determine the level of component redundancy in each subsystem and the quantity and location of spares to support the operation of the system. Simultaneously solving these two problems is a key contribution to the literature. The system design configuration is a reliability allocation problem, and the post-production support network composition is a spare optimization problem. We simultaneously solve these problems by extending the performancebased, multi-echelon, multi-item spares models, accounting for a series-parallel system of [55] and [68].

The structure of a PBC often requires an agreement between the buyer and the seller on multiple system performance metrics [42], [63]. Understanding these requirements is essential during the design process through the delivery and support of the system. Our model allows for evaluating multiple system performance metrics when changes are made to the system design or the complement of post-production resources.

The developed meta-heuristic is based on an evolutionary algorithm. Evolutionary algorithms are characterized by their population-based approach, iterative and adaptability evolution to improve candidate solutions over multiple generations based on the performance of the evolving population, and then choosing the best among them to create the next generation. The algorithm will terminate until the conditions are met; otherwise, it proceeds to the next generation. The output is the best solution(s) found during the evolutionary process as the final result [81]. It is worth mentioning that inside our algorithm, we are considering the optimization of the METRIC-based spares algorithm done by Nowicki et al. [68]. As a result, the proposed, evolutionary, meta-heuristic algorithm is an iterative, four-stage process that solves the model defined in section III-E. The stages and their relationships in an algorithmic flow chart are shown in Fig. 6. Stage 1 generates system designs or populates the model with possible initial solutions. Stage 2 calculates each design's profit, spares, reliability, availability, and logistics footprint so we can choose the best system designs. Stage 3 penalizes a design if it violates the buyer's performance requirements. This part of the algorithm agrees with the phase of evolutionary algorithms by penalizing the local search when it returns to a local minimum [82]. Stage 4 determines which designs are selected to compare against the next generated design set.

A. STAGE 1: GENERATE SYSTEM DESIGNS

Monte Carlo simulation generates *H* potential series-parallel system design configurations. Each design configuration is defined by the vector $x^h = [x_{j_i}^h]$ for $\forall i \in I$, $j_i \in J_i$, $h \in H$, where $x_{j_i} \in (0, 1)$. To determine if and where component redundancy is present, we define an inclusion vector $\Omega = [\Omega_{j_i}]$: $\Omega_{j_i} = Pr(x_{j_i} = 1)$ where $Pr(X_{j_i} = 1)$ is the probability of inclusion. A value ζ is randomly generated using a uniform distribution, U(0, 1). If $\zeta < \Omega_{j_i}$ then $x_{j_i}^h = 1$, otherwise set $x_{j_i}^h = 0$. The algorithm stops generating system designs when all initial inclusion probabilities are either zero or one, $\Omega_{j_i}^h = 1$ or $\Omega_{j_i}^h = 0 \forall i \in I$, $j_i \in J_i$, $h \in H$.

We initially assign an equal likelihood of inclusion to all potentially redundant components $\Omega_{j_i}^0 = 0 \quad \forall i \in I$, $j_i \in J_i$. Since the system configuration is series-parallel, there needs to be at least one component in each subsystem included, so we set $\Omega_{1_i}^0 = 1$. The equal likelihood of initial inclusion is based on a lack of knowledge of where and how many redundant components need to be included in the system design to meet the buyer-specified performance requirements. We apply the "Laplace principle of insufficient reason" [83] assigning an equal chance that a redundant component is included when knowledge is lacking.

B. STAGE 2: ANALYZE THE SYSTEM DESIGNS

This stage is crucial as it ensures the system design analysis and uses the computational efficiency of the metric-based spares algorithm. The system designs, x^h , generated in stage 1 are analyzed, and the values for $\Pi(\mathbf{x}^h, \mathbf{s}^h)$, $R(\mathbf{x}^h, \mathbf{s}^h)$, $C(\mathbf{x}^h, \mathbf{s}^h)$, $MTBF(\mathbf{x}^h, \mathbf{s}^h)$, $LF(\mathbf{x}^h, \mathbf{s}^h)$, and $A_0(x^h, \mathbf{s}^h)$. are calculated. Prior to determining these values, the quantity and location of spares, \mathbf{s}^h , need to be derived. To derive \mathbf{s}^h , we use the Multi-Echelon Technique for Recoverable Item Control (METRIC) approach [67] in the context of PBC [55] using



FIGURE 6. Algorithm flow chart.

computational efficiency improvements [68]. The result is a five-step process.

Step 1. Calculate $s_{j_i l_e}^0$ for each system design \mathbf{x}^h . First, calculate the expected backorder, $E[BO(d_{j_i l_e}(t))|, s_{j_i l_e} = 0]$ for j^{th} component in the i^{th} subsystem located in the l^{th} location in the e^{th} echelon of the post-production support network assuming there is no stock, $s_{j_i l_e} = 0$. If the $E[BO(d_{j_i l_e}(t))|s_{j_i l_e} = 0]] > EBO^{min}$, then set $s_{j_i l_e}^0 = \gamma$ $E[BO(d_{j_i l_e}(t)|s_{j_i l_e} = 0)]$, otherwise set $s_{j_i l_e}^0 = 0$. γ is the fraction of the $E[BO(d_{j_i l_e}(t)|s_{j_i l_e} = 0)]$ that is used to determine the initial number of component spares. This approach reduces the computations needed to reach an optimal multi-item, multi-echelon, inventory solution.

Step 2. Set the stock levels $s_{j_i l_e}$ equal to the initial stock levels from step 1, $s_{j_i l_e}^0$, and then calculate the expected backorders, $E[BO(d_{j_i l_e}(t)|s_{j_i l_e}^0)]$

Step 3. Increment the stock levels. The increments depend on the *EBO^{min}*.

If the $E[BO(d_{j_i l_0}(t) | s_{j_i l}^0] > EBO^{min}$ then increment by $\gamma E[BO(d_{j_i l_e}(t) | s_{j_i l_e}]$, otherwise increment by 1. So the $\Delta s_{j_i1} = s_{j_i1} + \gamma E[BO(d_{j_il_e}(t)|s_{j_il_e}] \text{ if } E[BO(d_{j_il_e}(t)|s_{j_il_e}^0] > EBO^{min} \text{ or } s_{j_il_e} + 1 \text{ otherwise. Therefore, the change in the expected backorder is } \Delta E[BO(d_{j_il_e}(t)|\Delta s_{j_il_e})] = E[BO(d_{j_il_e}(t)|\Delta s_{j_il_e})] - E[BO(d_{j_il_e}(t)|s_{j_il_e})] \forall i \in I, j_i \in J_i, l_e \in L_e, e \in E.$

Step 4. Choose the component, subassembly, location, and echelon combination (i, j_i, l_e, e) with the largest marginal profit $\Delta \Pi(x^h, s^h)$. Increment the stock levels for the chosen (i, j_i, l_e, e) by $\Delta s_{j_i l_e}$. The $\Delta \Pi(\mathbf{x}^h, \mathbf{s}^h) = \Delta R(\mathbf{x}^h, \mathbf{s}^h) - \Delta C(\mathbf{x}^h, \mathbf{s}^h)$, where the $\Delta C(\mathbf{x}^h, \mathbf{s}^h) = C(\mathbf{x}, \mathbf{s}) = \sum_e^{|E|} \sum_{l=1}^{|I_e|} \sum_{j=1}^{|I_i|} \sum_{j=1}^{|J_i|} c_{j_i l_e} + (\Delta s_{j_i l_e} + n_j x_{j_i l_e})$ from equation 2. $\Delta R(\mathbf{x}^h, \mathbf{s}^h)$ is derived from equation 6, leveraging the relationship between expected backorders and operational available is equation 3, $\Delta A_0 = 100 \prod_{L_E}^{|E|} \prod_{i=1}^{|I_i|} \{(1 - [E[BO(d_{j_i l_e}(t)|\Delta s_{j_i l_e})]]n_{l_{|E|}})\}^{n_{l_{|E|}}}$

Step 5. If the optimal profit is met, stop; otherwise, start again with step 2. Profit $\Pi(\mathbf{x}^h, \mathbf{s}^h)$ is convex, where the marginal profit $\Delta \Pi(\mathbf{x}^h, \mathbf{s}^h)$, decreases until zero. Then, it is concave, with a decreasing negative marginal profit. When applying our greedy algorithm, an optimal profit is reached

when the marginal profit value switches from positive to negative. the values of (x^h, s^h) that result in the optimal profit are then used to calculate $LF(\mathbf{x}^h, \mathbf{s}^h)$, and $MTBF(\mathbf{x}^h, \mathbf{s}^h)$.

C. STAGE 3: PENALIZE THE SYSTEM DESIGNS

The resulting supplier's profit, $\Pi(\mathbf{x}^h, \mathbf{s}^h)$, for each potential system design configuration (h) is penalized if it does not meet the buyer-specified $MTBF^T$ or LF^T contractually specified requirements. The penalty for violating the $MTBF^{T}$ or LF^T are:

$$P^{MTBF}(x^{h}) = \begin{cases} \frac{MTBF(x^{h})}{(\kappa \times MTBF^{T})} & \text{if } MTBF(x^{h}) \le MTBF^{T} \\ 1 & otherwise \end{cases}$$
(8)
$$P^{LF}(x^{h}, s^{h}) = \begin{cases} \frac{LF^{T}}{(\kappa \times (LF(x^{h}, s^{h})))} & \text{if } LF(x^{h}, s^{h}) \ge LF^{T} \\ 1 & otherwise \end{cases}$$
(9)

otherwise

In practice, PBCs may contain language suggesting the relative severity for not meeting contractually acceptable performance metrics (e.g., LF and MTBF). The factor κ represents the magnitude of the penalty for not meeting the LF and MTBF targets. Since κ is a constant, the penalties increase as the values move further from their targets. To penalize the competing, combined system design x^h and post-production support network s^h , the penalty metrics in Eq. 8 and Eq.9 are combined into a single penalty metric,

$$P(\mathbf{x}^h, \mathbf{s}^h) = P^{MTBF}(\mathbf{x}^h, \mathbf{s}^h) \times P^{LF}(\mathbf{x}^h, \mathbf{s}^h)$$
(10)

As specified in the model, the objective is to determine the combined system design x^h and post-production support network s^h , that maximizes the supplier's profit $\Pi(\mathbf{x}^h, \mathbf{s}^h)$ and meets the buyer-specified, performance targets, $MTBF(\mathbf{x}) >$ $MTBF^T$ and $LF(\mathbf{x}, \mathbf{s}) \leq LF^T$. This is accomplished by adjusting the profit of each competing design/postproduction support network combination by its penalty factor. The penalty-adjusted profit values of the competing system design x^h and their resulting post-production support network s^h are,

$$V(\mathbf{x}^{h}, \mathbf{s}^{h}) = \begin{cases} P(\mathbf{x}^{h}, \mathbf{s}^{h}) \times \Pi(\mathbf{x}^{h}, \mathbf{s}^{h}) & \text{if } \Pi(\mathbf{x}^{h}, \mathbf{s}^{h}) \ge 0\\ \Pi(\mathbf{x}^{h}, \mathbf{s}^{h}) / P(\mathbf{x}^{h}, \mathbf{s}^{h}) & \text{if } \Pi(\mathbf{x}^{h}, \mathbf{s}^{h}) < 0 \end{cases}$$
(11)

D. STAGE 4: SELECT SYSTEM DESIGN

First, the penalty-adjusted profit values of the competing system design \mathbf{x}^h and their resulting post-production support network are ordinally ranked, $V(\mathbf{x}^1, \mathbf{s}^1) \geq V(\mathbf{x}^2, \mathbf{s}^2) \geq$ $\dots V(\mathbf{x}^h, \mathbf{s}^h) \geq \dots V(\mathbf{x}^H, \mathbf{s}^H)$. We then identify a subset of the competing system designs with the largest penalty-adjusted profit values. The subset contains the largest $\omega|H|$, where ω is the fraction of the competing designs |H|. Next, we update the entries of the inclusion vector $\Omega'_{j_i} = \sum_{\rho=1}^{\omega|H|} x_{j_i l_e}^{(\rho)} / \omega|H|, \forall i \in I, j_i \in J_i, l_e \in L_e, e \in E.$ The updated inclusion vector, Ω , is then used as the initial conditions to generate additional system design configuration (stage 1). Our meta-heuristic algorithm (Fig.6) is consistent with the evolutionary strategy of a parent selection process, a reproduction strategy, and a substitution strategy, and its application to optimization algorithms [81]. The algorithm will stop when all the entries of the inclusion vector, Ω'_{i_i} , are either zero or one.

V. NUMERICAL STUDY

We provide a numerical study that illustrates the importance of simultaneously considering design and post-production support decisions in developing, implementing, and PBC. Fig. 7 shows the system design and post-production support system studied in the numerical example and the values used in the model (refer to the legends in Fig. 7). The system design is a series-parallel configuration with four subsystems in series, and each subsystem potentially has multiple redundant components. The post-production support system (i.e., postproduction support network) is a non-arboreal, three-echelon sustainment network with one central, two intermediate, and six field locations. Similar post-production support network designs can be found in [55], [68], and [26]. The constraints have targets for MTBF of at least 75 hours, while the LF is at most 10,000 lb.

The supplier revenue model used in this example is dependent on the availability achieved when operating the system design:

$$R[A_0(x, s)] = \begin{cases} 0 & \text{if } 0 \le A_0 < 0.70 \\ -60,000 + 300,000(A_0 - A_{min}) & \text{if } 0.70 \le A_0 < 0.80 \\ -30,000 + 416,666(A_0 - A_{penalty}) & \text{if } 0.80 \le A_0 < 0.95 \\ 45,000 + 100,000(A_0 - A_{accept}) & \text{if } 0.95 \le A_0 < 1.0 \end{cases}$$

$$(12)$$

We apply our analytical model to derive the solution shown in Fig. 8. The optimal system design and post-production support network results in a \$15,359.89 profit for the supplier with a LF of 5,904.50 lbs and an MTBF of 77.29 hours, satisfying the buyer-specified LF target of 10,000 lbs and MTFB target of 75 hours. There are two subsystems with one redundant component and two subsystems that have no redundancy reaching an availability of 0.9503.

The resulting system design vector \mathbf{x} is shown in Table 1. Table 2 displays how the inclusion vector Ω evolved with each model iteration.

Table 3 shows promising, feasible solutions that may be useful in the tradeoffs in buyer-supplier PBC negotiations. These feasible solutions, captured through the iterations of the algorithm and the convergent optimal solution, provide insights into the tradeoff space between MTBF, LF, and Profit.



Post-production Support Network

FIGURE 7. System Design and its Post-production Support Network.

TABLE 1. System design.

<i>x</i> ₁₁	<i>x</i> ₁₂	x_{1_3}	x_{1_4}	<i>x</i> ₂₁	x_{2_2}	<i>x</i> ₂₃	x_{2_4}	<i>x</i> ₃₁	x_{3_2}	<i>x</i> 33	x_{3_4}	<i>x</i> ₄₁	<i>x</i> ₄₂	x_{4_3}	<i>x</i> 44
1	0	1	0	1	0	0	0	1	0	0	0	1	0	0	1

TABLE 2. Changes in Ω vector per iteration.

Run	Ω_{1_1}	Ω_{1_2}	Ω_{1_3}	Ω_{1_4}	Ω_{2_1}	Ω_{2_2}	Ω_{2_3}	Ω_{2_4}	Ω_{3_1}	Ω_{3_2}	Ω_{3_3}	Ω_{3_4}	Ω_{4_1}	Ω_{4_2}	Ω_{4_3}	Ω_{4_4}
1	1	0.50	0.50	0.50	1	0.50	0.50	0.50	1	0.50	0.50	0.50	1	0.50	0.50	0.50
2	1	0.40	0.50	0.35	1	0.20	0.30	0.45	1	0.45	0.30	0.45	1	0.15	0.25	0.40
3	1	0.20	0.35	0.45	1	0.20	0.25	0.30	1	0.30	0.15	0.35	1	0.05	0.15	0.30
4	1	0.10	0.40	0.20	1	0.10	0.10	0.30	1	0.30	0.20	0.30	1	0.05	0.20	0.45
5	1	0.10	0.45	0.15	1	0.10	0.05	0.20	1	0.40	0.10	0.20	1	0.05	0.15	0.55
6	1	0	0.40	0.10	1	0	0	0.20	1	0.45	0.05	0.05	1	0	0.05	0.85
7	1	0	0.65	0	1	0	0	0	1	0.30	0	0.05	1	0	0	1
8	1	0	1	0	1	0	0	0	1	0	0	0	1	0	0	1

We perform a comparative experiment showing that better decisions are made when simultaneously considering system design and the post-production support network rather than making these decisions sequentially or often independently. Tables 4 and 5 show the results of the comparative experiment, providing the details supporting the use of our proposed simultaneous analysis rather than the traditional sequential approach.

Solving the systems separately results in no satisfaction of the contractual constraints. We perform a set of experiments by first solving the system design with no redundancy, then the post-production support network optimization. We add redundancy to each subsystem one by one until we find a feasible solution that satisfies the contractual constraints of MTBF and LF. Table 4 shows the system designs of the experiments conducted, MTBF, LF, and Cost. While Table 5



System Design

Post-production Support Network

FIGURE 8. System design and post-production support network optimization.

illustrates for each run the LF, Cost, Revenue, and availability of the post-production support network optimization. Finally, the last three columns of Table 5 show whether solving the systems separately the contractual constraints are met. Up to experiment k, both contractual constraints are met, and the system design adds two redundant components to subsystems 1 and 2 with a MTBF of 76.17 hours and a total LF of 6,131.75 lb. Both systems result in a profit of \$ 10,388.83. In addition, experiment l has a feasible solution. However, these runs have lower profits than solving the systems simultaneously with the proposed model.

VI. MANAGERIAL INSIGHTS

PBC is used in both public and private sectors, generating buyer value and supplier-buyer profitability. PBC is mostly used in systems where the cost of sustaining the system exceeds its research, development, and production costs. The incentive scheme is for the supplier to deliver performance outcomes for long-term contracts and increase the return on investment. The business implications for PBC are significant. Our current research focuses on evaluating the system design and its support network, which have the potential to significantly enhance the success and proliferation of PBC across multiple industry sectors. This research provides valuable insights, a roadmap, and supporting analytical models to further PBC across many industry sectors, particularly in evaluating the inherent trade-offs between system design and the network of support resources necessary to maintain these systems [84] The untapped potential of PBC offers firms (buyers) purchasing systems and firms (suppliers) designing, producing, and supporting these systems the promise of improved system performance and outcomes, cost efficiency, risk management through innovations, long-term relationships between suppliers and buyers, and most importantly, it has the power to enhance market competitiveness, driving the industry forward.

In many industries, there are often two contracts between the buyer and supplier of a system and another between the buyer and the supplier for the support network to maintain system performance and outcome, leading to suboptimal cost efficiency. With the present research, there is a promising potential to have only one contract that specifies the system design and its operating performance and outcome, thereby significantly improving cost efficiency.

The U.S. DoD has successfully employed PBC to design and post-production support large-scale, complex, repairable systems, such as aircraft. These contracts, which are often separate, could be better understood and managed by buyers and suppliers. This understanding could be contractually formalized, leading to improved performance and higher quality of service. The unique benefits of PBC in this context are evident, as it has the potential to significantly enhance

TABLE 3. Other feasible solutions to the numerical example.

Nu	mber of compor					
Subsystem 1	Subsystem 2	Subsystem 3	Subsystem 4	MTBF (hrs)	LF (lb)	Profit (\$)
1	1	2	2	78.45	6,285.25	14,505.12
2	2	2	1	78.45	5,929.50	13,540.52
3	1	1	2	79.42	6,415.75	12,828.55
1	2	1	2	78.45	6,310.25	12,332.62
3	2	2	1	80.64	6,402.75	10,781.00

TABLE 4. Experiments of the system design without optimization.

	System Design										
	Number of	components per	r subsystem								
Exp.	Subsystem 1	Subsystem 2	Subsystem 3	Subsystem 4	MTBF (hrs)	LF (lb)	Cost (\$)				
a	1	1	1	1	62.57	1,092.25	6,375				
b	2	1	1	1	66.53	1,298.50	7,625				
с	3	1	1	1	68.10	1,504.75	8,875				
d	4	1	1	1	68.98	1,711.00	10,125				
e	1	2	1	1	67.38	1,347.25	8,625				
f	2	2	1	1	72.00	1,553.50	9,875				
g	3	2	1	1	73.84	1,759.75	11,125				
h	4	2	1	1	74.87	1,966.00	12,375				
i	1	3	1	1	69.32	1,602.25	10,875				
j	1	3	1	1	74.22	1,808.50	12,125				
k	3	3	1	1	76.17	2,014.75	13,375				
1	4	3	1	1	77.27	2,221.00	14,625				

TABLE 5. Experiments of the post-production support network optimization after the system design decision, the total cost of both systems and the validity of the contractual constraints.

	Pos	t-productio	on support netw	ork	Both	Cons	straints
Exp.	LF (lb)	Cost (\$)	R[A(x,s)](\$)	A_0	Profit (\$)	$MTBF \ge 75$	$LF \leq 10,000$
a	2,814	13,775	45,394.50	0.9539	25,244.50	×	√
b	3,107	15,400	45,104.40	0.9510	22,079.40	×	\checkmark
с	3,403	17,025	45,007.54	0.9501	19,107.54	X	√
d	3,696	18,650	45,234.16	0.9523	16,459.16	X	√
e	3,164	16,025	45,299.05	0.9530	20,649.05	×	√
f	3,462	17,700	45,034.89	0.9503	17,459.89	X	√
g	3,783	19,400	45,442.54	0.9544	14,917.54	X	√
h	3,043	20,925	45,212.19	0.9521	11,912.19	×	√
i	3,519	18,325	45,429.89	0.9543	16,229.89	X	√
j	8,812	19,950	45,193.07	0.9519	13,118.07	×	\checkmark
k	4,117	21,600	45,363.83	0.9536	10,388.83	√	√
1	4,385	23,125	45,016.18	0.9501	7,266.18	\checkmark	\checkmark

the success and proliferation of PBC across multiple industry sectors.

PBC contracts that simultaneously consider design and support could lead to significant innovations. For instance, an insurance company (buyer) might incentivize a healthcare provider (supplier) to reduce hospital readmission rates. To achieve this, they could develop (design) new post-discharge care programs, such as remote patient monitoring systems (operation and support), which use wearable technology to track patient health data and provide realtime feedback. This innovative approach improves patient care (performance and outcome) and introduces cutting-edge technology and processes (innovation), driving the industry forward [84]. Energy companies use PBC to maintain and operate power plants [85]. For example, a power plant operator might have a PBC contract where payment depends on plant efficiency and environmental compliance. This performance-based payment structure motivates the operator to optimize operations, reduce emissions, and improve energy efficiency, leading to lower operational costs and better environmental outcomes. A PBC contract with the power plant designer and operator could improve performance and encourage suppliers to invest in innovation. Power plant operators might adopt innovative technologies such as AI-based smart grids to optimize electricity distribution and consumption to meet energy efficiency and emissions performance targets. The industrial benefits of this research are as follows:

- Two complex systems are analyzed together. The first is a post-production support network, a non-arboreal muti-echelon sustainment network, and the second is its system design, a series-parallel configuration.
- The model maximizes a large-scale, complex system's profit margin while considering both the design and post-production support network.
- The model considers performance metrics valid to the current requirements of operability and sustainability.
- The model is constrained by minimum reliability, measured as the MTBF, and a maximum LF, measured in pounds. These contractual buyer-specified requirements can be modified according to the agreed system's availability and environmental goals.
- The demand rate is the component in the subassembly failure rate multiplied by the number of redundant components in the subassembly.

VII. CONCLUSION, DISCUSSION, AND FUTURE RESEARCH

This research aims to introduce an analytical model that simultaneously considers design and post-production support decisions in developing, implementing, and managing PBC. We are the first to create an analytical model that concurrently considers design and post-production support decisions in the context of PBC. We use the redundancy allocation problem to represent a design decision, and the post-production support network decisions are the location and quantity of spares and their logistics footprint. The post-production support network is a non-arboreal, multi-echelon sustainment network, and the design is a series-parallel configuration. The model is constrained by buyer-specified, minimum reliability, measured as the MTBF, and a maximum LF, measured in pounds. A numerical example was solved to test the model and to highlight the effectiveness of the proposed meta-heuristic by showing different feasible optimal solutions. The solution demonstrates the economic importance of system engineers, contract personnel, and program managers in understanding the inherent trade-off space connecting the design and support of a system.

The most important limitation is that detailed data about failure rates, repair time, logistics lead times, and logistics footprint is required. Gathering and maintaining this data is complex and costly. In addition, solving large-scale systems with many items and multiple echelons can make real-time or near-real-time applications challenging. Our model focuses primarily on holding costs, shortage costs, logistics footprint, and availability performance measures, such as the MTBF. It may not capture other relevant costs like transportation, administration, or downtime.

This research provides the foundation to further study collaboration between buyers and sellers, specifically how an optimal price is determined for a performance-based contract (e.g., game-theoretic models). With PBC, suppliers are incentivized to make improvements in the system's design, as long as these investments produce future rewards over the contract that exceed the investment decisions. Theoretical research is needed to understand further foundational theories such as transactional cost economics, management control theory, and agency theory that influence buyer-seller behavior in PBC's context. These incentives may spur design innovation; innovative designs that improve system reliability result in fewer failures or demands on post-production resources. The result is a more sustainable post-production support system. Consequently, future theoretical and empirical research is needed to study how PBC influences system design and post-production support innovation. Additionally, PCB impacts sustainability when management considers system designs and post-production support decisions simultaneously. Several research topics are worthy of further exploration. Resilience is one of those topics. For instance, it would be worth analyzing the impact of failures in the post-production support network regarding availability and investment needed to restore it to a satisfactory availability level [26]. It is even more important to evaluate the long-term economic effects of disruptive events, such as the COVID-19 pandemic. There is undoubtedly a linkage between design and post-production support in determining a system's resiliency. There is an opportunity to study the linkage further.

A further study could assess the implementation of new technology for spare parts replacement, such as autonomation inspection, to avoid failures [86]. Imperfect replacement or periodic inspection schemes and other performance metrics can be considered in future research, such as condition-based maintenance [30]. Our study considers only one supplier; however, suppliers often depend on other suppliers and subsuppliers. A study evaluating these supplier dependencies and how PBC is affected would be worth analyzing [87]. A natural progression of this work is that environmental responsibility could produce interesting findings that account more for benefits for the supplier. For example, the less logistics footprint of the system, the more economic benefits there are for the supplier. Also, involving competition among suppliers for better performance metrics would be an important avenue to investigate [88]. It would be worth evaluating our model in a real-world case study to evaluate all the managerial implications and complexities of systems design.

REFERENCES

- A. Sols, D. Nowick, and D. Verma, "Defining the fundamental framework of an effective performance-based logistics (PBL) contract," *Eng. Manage. J.*, vol. 19, no. 2, pp. 40–50, Jun. 2007.
- [2] J. Boyce and A. Banghart, "Performance based logistics and project proof point," *Defense AT L, Product Support Issue*, vol. 41, no. 2, pp. 26–30, 2012.
- [3] W. McEwan and J. Butterfield, "A digital methodology for the design process of aerospace assemblies with sustainable composite processes & manufacture," *AIP Conf. Proc.*, vol. 1353, no. 1, pp. 1683–1688, 2011.
- [4] C. J. Hockley, J. C. Smith, and L. J. Lacey, "Contracting for availability and capability in the defence environment," in *Complex Engineering Service Systems*. London, U.K.: Springer, 2011, pp. 237–256.
- [5] J. S. Gansler, W. Lucyshyn, and C. Vorhis, "Performance-based services acquisition," Center Public Policy Private Enterpise, Univ. Maryland, Tech. Rep. UMD-CM-11-005, 2011. [Online]. Available: https://dair.nps.edu/bitstream/123456789/2479/1/UMD-CM-11-005.pdf

- [6] P. Sandborn, A. Kashani-Pour, N. Goudarzi, and X. Lei, "Outcome-based contracts—Towards concurrently designing products and contracts," *Proc. CIRP*, vol. 59, pp. 8–13, Jan. 2017.
- [7] W. S. Randall, T. L. Pohlen, and J. B. Hanna, "Evolving a theory of performance-based logistics using insights from service dominant logic," *J. Bus. Logistics*, vol. 31, no. 2, pp. 35–61, Sep. 2010.
- [8] P. Hypko, M. Tilebein, and R. Gleich, "Benefits and uncertainties of performance-based contracting in manufacturing industries: An agency theory perspective," *J. Service Manage.*, vol. 21, no. 4, pp. 460–489, Aug. 2010.
- [9] K. Selviaridis and F. Wynstra, "Performance-based contracting: A literature review and future research directions," *Int. J. Prod. Res.*, vol. 53, no. 12, pp. 3505–3540, Jun. 2015.
- [10] J. M. de la Garza and J. L. Arcella, "Current performance-based maintenance methods to improve Virginia highways: Comparative analysis," *Transp. Res. Rec., J. Transp. Res. Board*, vol. 2361, no. 1, pp. 35–43, Jan. 2013.
- [11] N. Radović, K. Mirković, M. Šešlija, and I. Peško, "Output and performance based road maintenance contracting—Case study Serbia," *Tech. Gazette*, vol. 21, no. 3, pp. 681–688, 2014.
- [12] A. Gajurel, Performance-Based Contracts for Road Projects. India: Springer, 2014.
- [13] S. M. Famurewa, U. Juntti, and U. Kumar, "Performance based railway infrastructure maintenance: Towards achieving maintenance objectives," in *Proc. Int. Conf. Maintenance Perform. Meas. Manage.* Luleå, Sweden: Luleå Tekniska Universitet, 2011, pp. 233–240.
- [14] M. C. D. Santos, J. M. Pinto, and P. V. Matos, "Performance-based contracting of urban transport operation services: Evidence from Porto's light-rail," *Case Stud. Transp. Policy*, vol. 16, Jun. 2024, Art. no. 101193.
- [15] W. Zeng, M. Cros, K. D. Wright, and D. S. Shepard, "Impact of performance-based financing on primary health care services in Haiti," *Health Policy Planning*, vol. 28, no. 6, pp. 596–605, Sep. 2013.
- [16] J. Wang, X. Zhao, and X. Guo, "Optimizing wind turbine's maintenance policies under performance-based contract," *Renew. Energy*, vol. 135, pp. 626–634, May 2019.
- [17] A. D. Papalexopoulos and P. E. Andrianesis, "Performance-based pricing of frequency regulation in electricity markets," *IEEE Trans. Power Syst.*, vol. 29, no. 1, pp. 441–449, Jan. 2014.
- [18] L. Kratz and G. Diaz, "Affordable logistics: Are we there yet," *Defense AT L*, vol. 41, no. 2, pp. 39–43, 2012.
- [19] S.-H. Kim, M. A. Cohen, S. Netessine, and S. Veeraraghavan, "Contracting for infrequent restoration and recovery of mission-critical systems," *Manage. Sci.*, vol. 56, no. 9, pp. 1551–1567, Sep. 2010.
- [20] PBL Guidebook: A Guide to Developing Performance-based Arrangements, U.S. Dept. Defense, Washington, DC, USA, 2014.
- [21] J. A. Guajardo, M. A. Cohen, S.-H. Kim, and S. Netessine, "Impact of performance-based contracting on product reliability: An empirical analysis," *Manage. Sci.*, vol. 58, no. 5, pp. 961–979, May 2012. [Online]. Available: http://www.ssrn.com/abstract=1807049
- [22] S.-H. Kim, M. A. Cohen, and S. Netessine, "Performance contracting in after-sales service supply chains," *Manage. Sci.*, vol. 53, no. 12, pp. 1843–1858, Dec. 2007.
- [23] Q. Hu, J. E. Boylan, H. Chen, and A. Labib, "OR in spare parts management: A review," *Eur. J. Oper. Res.*, vol. 266, no. 2, pp. 395–414, Apr. 2018, doi: 10.1016/j.ejor.2017.07.058.
- [24] M. Kalchschmidt, G. Zotteri, and R. Verganti, "Inventory management in a multi-echelon spare parts supply chain," *Int. J. Prod. Econ.*, vols. 81–82, pp. 397–413, Jan. 2003.
- [25] S. Rahimi-Ghahroodi, A. Al Hanbali, I. M. H. Vliegen, and M. A. Cohen, "Joint optimization of spare parts inventory and service engineers staffing with full backlogging," *Int. J. Prod. Econ.*, vol. 212, pp. 39–50, Jun. 2019, doi: 10.1016/j.ijpe.2019.02.007.
- [26] A. Zavala, D. Nowicki, and J. E. Ramirez-Marquez, "Quantitative metrics to analyze supply chain resilience and associated costs," *Proc. Inst. Mech. Eng. O, J. Risk Rel.*, vol. 233, no. 2, pp. 186–199, Apr. 2019. [Online]. Available: http://journals.sagepub.com/doi/10.1177/ 1748006X18766738
- [27] F. Costantino, G. Di Gravio, and M. Tronci, "Multi-echelon, multiindenture spare parts inventory control subject to system availability and budget constraints," *Rel. Eng. Syst. Saf.*, vol. 119, pp. 95–101, Nov. 2013. [Online]. Available: http://www.sciencedirect.com/science/ article/pii/S0951832013001312

- [28] Z. S. Hua, B. Zhang, J. Yang, and D. S. Tan, "A new approach of forecasting intermittent demand for spare parts inventories in the process industries," *J. Oper. Res. Soc.*, vol. 58, no. 1, pp. 52–61, Jan. 2007. [Online]. Available: https://www.tandfonline.com/doi/full/ 10.1057/palgrave.jors.2602119
- [29] F. Alqahtani, K. Selviaridis, and M. Stevenson, "The effectiveness of performance-based contracting in the defence sector: A systematic literature review," *J. Purchasing Supply Manage.*, vol. 29, no. 5, Dec. 2023, Art. no. 100877.
- [30] Y. Wang, W. Gao, X. Li, and Y. Liu, "Joint optimization of performancebased contracting, condition-based maintenance and spare parts inventory for degrading production systems," *Rel. Eng. Syst. Saf.*, vol. 243, Mar. 2024, Art. no. 109845.
- [31] D. W. Coit and A. E. Smith, "Reliability optimization of series-parallel systems using a genetic algorithm," *IEEE Trans. Rel.*, vol. 45, no. 2, pp. 254–260, Jun. 1996.
- [32] W. J. Kennedy, J. Wayne Patterson, and L. D. Fredendall, "An overview of recent literature on spare parts inventories," *Int. J. Prod. Econ.*, vol. 76, no. 2, pp. 201–215, Mar. 2002.
- [33] A. Bacchetti and N. Saccani, "Spare parts classification and demand forecasting for stock control: Investigating the gap between research and practice," *Omega*, vol. 40, no. 6, pp. 722–737, Dec. 2012, doi: 10.1016/j.omega.2011.06.008.
- [34] S. Devi, H. Garg, and D. Garg, "A review of redundancy allocation problem for two decades: Bibliometrics and future directions," *Artif. Intell. Rev.*, vol. 56, no. 8, pp. 7457–7548, Aug. 2023.
- [35] M. A. Ardakan, M. Sima, A. Z. Hamadani, and D. W. Coit, "A novel strategy for redundant components in reliability-redundancy allocation problems," *IIE Trans.*, vol. 48, no. 11, pp. 1043–1057, Nov. 2016.
- [36] N. H. Zavieh, M. A. Ardakan, and H. Davari-Ardakani, "A new model for the reliability-redundancy allocation problem under the Kmixed redundancy strategy," *J. Stat. Comput. Simul.*, vol. 92, no. 17, pp. 3542–3560, Nov. 2022.
- [37] S. C. Sup and C. Y. Kwon, "Branch-and-bound redundancy optimization for a series system with multiple-choice constraints," *IEEE Trans. Rel.*, vol. 48, no. 2, pp. 108–117, Jun. 1999.
- [38] M. Sharifi and S. Taghipour, "Redundancy allocation problem of a multistate system with binary-state continuous performance level components," *Expert Syst. Appl.*, vol. 200, Aug. 2022, Art. no. 117161.
- [39] A. Chambari, P. Azimi, and A. A. Najafi, "A bi-objective simulationbased optimization algorithm for redundancy allocation problem in seriesparallel systems," *Expert Syst. Appl.*, vol. 173, Jul. 2021, Art. no. 114745.
- [40] K.-H. Chang and P.-Y. Kuo, "An efficient simulation optimization method for the generalized redundancy allocation problem," *Eur. J. Oper. Res.*, vol. 265, no. 3, pp. 1094–1101, Mar. 2018, doi: 10.1016/j.ejor.2017.08.049.
- [41] P. Scarf, A. Syntetos, and R. Teunter, "Joint maintenance and spare-parts inventory models: A review and discussion of practical stock-keeping rules," *IMA J. Manage. Math.*, vol. 35, no. 1, pp. 83–109, Dec. 2023.
- [42] M. Hemmati, M. Amiri, and M. Zandieh, "Optimization redundancy allocation problem with nonexponential repairable components using simulation approach and artificial neural network," *Qual. Rel. Eng. Int.*, vol. 34, no. 3, pp. 278–297, Apr. 2018, doi: 10.1002/qre.2249.
- [43] J. J. Hadel and P. B. Lakey, "A customer-oriented approach to optimizing reliability-allocation within a set of weapon-system requirements," in *Proc. Annu. Rel. Maintainability Symp.*, 1995, pp. 96–101.
- [44] T. R. Edison and A. Murphy, "A new look at enablers and barriers to performance based life cycle product support (PBL) implementation," Defense Acquisition Univ., Fort Belvoir, VA, USA, Tech. Rep. 4, 2012.
- [45] S. Geary and K. Vitasek, "Performance-based logistics: A contractor's guide to life cycle product support management," Supply Chain Vis., Bellevue, WA, USA, 2008.
- [46] D. Berkowitz, J. N. D. Gupta, J. T. Simpson, and J. B. McWilliams, "Defining and implementing performance-based logistics in government," *Defense Acquisition Rev. J.*, vol. 11, no. 3, pp. 255–267, 2005.
- [47] D. Nowicki, W. S. Randall, and A. Gorod, "A framework for performance based logistics: A system of systems approach," in *Proc. Int. Congr. Ultra Modern Telecommun. Control Syst.*, Oct. 2010, pp. 681–692.
- [48] S. Prokesch, "How GE teaches teams to lead change," *IEEE Eng. Manag. Rev.*, vol. 40, no. 4, pp. 31–41, Dec. 2012.
- [49] A. Sols, D. Nowicki, and D. Verma, "N-Dimensional effectiveness metriccompensating reward scheme in performance-based logistics contracts," *Syst. Eng.*, vol. 11, no. 2, pp. 93–106, Jun. 2008.

- [50] S.-H. Kim, M. A. Cohen, and S. Netessine, "Reliability or inventory? An analysis of performance-based contracts for product support services," in *Handbook of Information Exchange in Supply Chain Management.* Switzerland: Springer, 2017, pp. 65–88. [Online]. Available: http://link.springer.com/10.1007/978-3-319-32441-8_4
- [51] S.-H. Kim, J. A. Guajardo, and S. Netessine, "Performance-based contracting: Past, present, and future," in *Creating Values With Operations* and Analytics (Springer Series in Supply Chain Management), vol. 19, H. Lee, R. Ernst, A. Huchzermeier, and S. Cui, Eds., Cham, Switzerland: Springer, 2022, pp. 85–103, doi: 10.1007/978-3-031-08871-1_6.
- [52] K. Selviaridis and A. Norrman, "Performance-based contracting for advanced logistics services," *Int. J. Phys. Distrib. Logistics Manage.*, vol. 45, no. 6, pp. 592–617, Jul. 2015. [Online]. Available: http://www.emeraldinsight.com/doi/10.1108/IJPDLM-11-2014-0267
- [53] D. Nowicki, B. Sauser, W. Randall, and R. Lusch, "Service-dominant logic and performance-based contracting: A systems thinking perspective," *Service Sci.*, vol. 10, no. 1, pp. 12–24, Mar. 2018. [Online]. Available: http://pubsonline.informs.org/doi/10.1287/serv.2017.0185
- [54] N. Kontrec, S. Panić, M. Petrović, and H. Milošević, "A stochastic model for estimation of repair rate for system operating under performance based logistics," *Maintenance Rel.*, vol. 20, no. 1, pp. 68–72, Mar. 2018.
- [55] D. Nowicki, U. D. Kumar, H. J. Steudel, and D. Verma, "Spares provisioning under performance-based logistics contract: Profit-centric approach," J. Oper. Res. Soc., vol. 59, no. 3, pp. 342–352, Mar. 2008.
- [56] W. S. Randall, D. R. Nowicki, G. Deshpande, and R. F. Lusch, "Converting knowledge into value," *Int. J. Phys. Distrib. Logistics Manag.*, vol. 44, nos. 8–9, pp. 655–670, Sep. 2014. [Online]. Available: https://www.emerald.com/insight/content/doi/10.1108/IJPDLM-08-2013-0223/full/html
- [57] W. S. Randall, D. R. Nowicki, and T. G. Hawkins, "Explaining the effectiveness of performance-based logistics: A quantitative examination," *Int. J. Logistics Manage.*, vol. 22, no. 3, pp. 324–348, Nov. 2011. [Online]. Available: http://www.emeraldinsight.com/ doi/10.1108/09574091111181354
- [58] J. van Strien, C. J. Gelderman, and J. Semeijn, "Performance-based contracting in military supply chains and the willingness to bear risks," *J. Defense Anal. Logistics*, vol. 3, no. 1, pp. 83–107, Jun. 2019.
- [59] A. H. Glas, C. Raithel, and M. Essig, "Risk perception in performance based contracts and the influence of experience," *Int. J. Productiv. Perform. Manage.*, vol. 68, no. 6, pp. 1078–1101, Jul. 2019.
- [60] A. H. Glas, F. U. Henne, and M. Essig, "Missing performance management and measurement aspects in performance-based contracting: A systematic process-based literature analysis of an astonishing research gap," *Int. J. Oper. Prod. Manage.*, vol. 38, no. 11, pp. 2062–2095, Nov. 2018.
- [61] P. Patra, U. D. Kumar, D. R. Nowicki, and W. S. Randall, "Effective management of performance-based contracts for sustainment dominant systems," *Int. J. Prod. Econ.*, vol. 208, pp. 369–382, Feb. 2019, doi: 10.1016/j.ijpe.2018.11.025.
- [62] Y. Tang, Y. Chen, D. Arditi, and F. Meng, "Effects of the general contractor's governance capabilities and project goals on the organizational arrangement of subcontracting," *IEEE Trans. Eng. Manag.*, vol. 70, no. 5, pp. 1724–1737, May 2023.
- [63] Z. Huang, Q. Gao, Y. Ge, and B. Hao, "Optimization of spares inventory based on supply availability," in *Proc. IEEE Prognostics Syst. Health Manage. Conf.*, May 2012, pp. 1–4.
- [64] D. W. Coit and J. C. Liu, "System reliability optimization with k-out-ofn subsystems," *Int. J. Rel., Quality Saf. Eng.*, vol. 7, no. 2, pp. 129–142, Jun. 2000.
- [65] S. C. Graves, "A multi-echelon inventory model for a repairable item with one-for-one replenishment," *Manage. Sci.*, vol. 31, no. 10, pp. 1247–1256, Oct. 1985.
- [66] C. C. Sherbrooke, "VARI-METRIC: Improved approximations for multiindenture, multi-echelon availability models," *Oper. Res.*, vol. 34, no. 2, pp. 311–319, Apr. 1986.
- [67] C. Sherbrooke, Optimal Inventory Modeling of Systems (International Series in Operations Research & Management Science), vol. 72. Boston, MA, USA: Springer, 2004. [Online]. Available: http://link.springer.com/ 10.1007/b109856
- [68] D. R. Nowicki, W. S. Randall, and J. E. Ramirez-Marquez, "Improving the computational efficiency of metric-based spares algorithms," *Eur. J. Oper. Res.*, vol. 219, no. 2, pp. 324–334, Jun. 2012, doi: 10.1016/j.ejor.2011.12.033.

- [69] L. Kratz, J. Beck, D. Verma, and T. Parry. (2003). Designing and Assessing Supportability in DOD Weapon Systems: A Guide to Increased Reliability and Reduced Logistics Footprint. [Online]. Available: http://www.dtic.mil/dtic/tr/fulltext/u2/a606329.pdf
- [70] Y. Asiedu and P. Gu, "Product life cycle cost analysis: State of the art review," Int. J. Prod. Res., vol. 36, no. 4, pp. 883–908, Apr. 1998. [Online]. Available: http://www.tandfonline.com/doi/abs/ 10.1080/002075498193444
- [71] B. Fabrycky, Systems Engineering and Analysis. Upper Saddle River, NJ, USA: Prentice-Hall, 2014.
- [72] N. Faber, R. B. M. de Koster, and S. L. van de Velde, "Linking warehouse complexity to warehouse planning and control structure: An exploratory study of the use of warehouse management information systems," *Int. J. Phys. Distrib. Logistics Manage.*, vol. 32, no. 5, pp. 381–395, Jun. 2002.
- [73] K. H. Wathne and J. B. Heide, "Relationship governance in a supply chain network," J. Marketing, vol. 68, no. 1, pp. 73–89, Jan. 2004. [Online]. Available: http://journals.ama.org/doi/abs/10.1509/jmkg.68.1.73.24037
- [74] B. D. Williams and T. Tokar, "A review of inventory management research in major logistics journals," *Int. J. Logistics Manage.*, vol. 19, no. 2, pp. 212–232, Aug. 2008. [Online]. Available: http://www.emeraldinsight.com/doi/10.1108/09574090810895960
- [75] U. D. Kumar, D. Nowicki, J. E. Ramìrez-Márquez, and D. Verma, "A goal programming model for optimizing reliability, maintainability and supportability under performance based logistics," *Int. J. Rel., Qual. Saf. Eng.*, vol. 14, no. 3, pp. 251–261, Jun. 2007.
- [76] B. Sundarakani, R. de Souza, M. Goh, S. M. Wagner, and S. Manikandan, "Modeling carbon footprints across the supply chain," *Int. J. Prod. Econ.*, vol. 128, no. 1, pp. 43–50, Nov. 2010, doi: 10.1016/j.ijpe.2010.01. 018.
- [77] V. K. Manupati, S. J. Jedidah, S. Gupta, A. Bhandari, and M. Ramkumar, "Optimization of a multi-echelon sustainable production-distribution supply chain system with lead time consideration under carbon emission policies," *Comput. Ind. Eng.*, vol. 135, pp. 1312–1323, Sep. 2019, doi: 10.1016/j.cie.2018.10.010.
- [78] F. Ye, G. Huang, Y. Zhan, and Y. Li, "Factors mediating and moderating the relationships between green practice and environmental performance: Buyer–supplier relation and institutional context," *IEEE Trans. Eng. Manag.*, vol. 70, no. 1, pp. 142–155, Jan. 2023.
- [79] R. E. Brown and J. J. Burke, "Managing the risk of performance based rates," *IEEE Trans. Power Syst.*, vol. 15, no. 2, pp. 893–898, May 2000.
- [80] X. Qin, Z.-Z. Jiang, M. Sun, L. Tang, and X. Liu, "Repairable spare parts provisioning for multiregional expanding fleets of equipment under performance-based contracting," *Omega*, vol. 102, Jul. 2021, Art. no. 102328.
- [81] Z. Ma, G. Wu, P. N. Suganthan, A. Song, and Q. Luo, "Performance assessment and exhaustive listing of 500+ nature-inspired metaheuristic algorithms," *Swarm Evol. Comput.*, vol. 77, Mar. 2023, Art. no. 101248, doi: 10.1016/j.swevo.2023.101248.
- [82] M. Gendreau and J.- Y. Potvin, *Handbook of Metaheuristics*, vol. 2. New York, NY, USA: Springer, 2010, doi: 10.1007/978-1-4419-1665-5.
- [83] D. Janzing, "Causal versions of maximum entropy and principle of insufficient reason," J. Causal Inference, vol. 9, no. 1, pp. 285–301, Dec. 2021.
- [84] M. M. Rakers, H. J. A. van Os, K. Recourt, G. Mosis, N. H. Chavannes, and J. N. Struijs, "Perceived barriers and facilitators of structural reimbursement for remote patient monitoring, an exploratory qualitative study," *Health Policy Technol.*, vol. 12, no. 1, Mar. 2023, Art. no. 100718.
- [85] D. Kirli, B. Couraud, V. Robu, M. Salgado-Bravo, S. Norbu, M. Andoni, I. Antonopoulos, M. Negrete-Pincetic, D. Flynn, and A. Kiprakis, "Smart contracts in energy systems: A systematic review of fundamental approaches and implementations," *Renew. Sustain. Energy Rev.*, vol. 158, Apr. 2022, Art. no. 112013.
- [86] B. K. Dey, B. Sarkar, and H. Seok, "Cost-effective smart autonomation policy for a hybrid manufacturing-remanufacturing," *Comput. Ind. Eng.*, vol. 162, Dec. 2021, Art. no. 107758.
- [87] A. Nikulina and F. Wynstra, "Understanding supplier motivation to engage in multiparty performance-based contracts: The lens of expectancy theory," J. Purchasing Supply Manage., vol. 28, no. 2, Mar. 2022, Art. no. 100746.
- [88] Y. Zhou, Y. Zhang, M. I. M. Wahab, and M. Goh, "Channel leadership and performance for a closed-loop supply chain considering competition," *Transp. Res. E, Logistics Transp. Rev.*, vol. 175, Jul. 2023, Art. no. 103151.



ARACELI ZAVALA received the bachelor's degree in industrial engineering from Instituto Tecnológico de Morelia, the Master of Science degree in quality and productivity from Tecnológico de Monterrey, and the Ph.D. degree in engineering management from the Stevens Institute of Technology. Since 2005, she has been with the Industrial Engineering Department, Tecnológico de Monterrey, Campus Guadalajara, as a full-time Professor. She has been a Consultant for several

small and big companies in Mexico. She is a member of the National System of Researchers of CONAHCYT in Mexico at level I.



DAVID NOWICKI received the bachelor's degree in industrial and systems engineering from the University of Wisconsin–Madison, the master's degree in industrial and systems engineering from Virginia Tech, and the Ph.D. degree in industrial and systems engineering from the University of Wisconsin–Madison.

He is currently a Professor with the University of North Texas and the Director of the G. Brint Ryan College of Business, Center for Logistics

and Supply Chain Management. He holds a joint appointment with the G. Brint Ryan College of Business, the Logistics and Operations Management Department, the College of Engineering, and the Mechanical Engineering Department. His research applies advanced analytical techniques to solve logistics and supply chain management problems in a systems engineering context. His research interests include performancebased contracts, supply chain management, resiliency and risk, game theory, multi-objective optimization, reliability theory, and inventory optimization. He has been awarded over \$7.7 million in competitive research grants as Principal or Co-Principal Investigator with thirty-five (35) funded proposals. He consistently publishes in top-tier research journals with 35 publications 1,706 citations, an H-index of 20, and an i-10 index of 24. He focuses on applied research, leveraging his two decades of industry experience, holding executive positions at i2 Technologies and the TFD Group.



JOSE EMMANUEL RAMIREZ-MARQUEZ is currently the Director of the Enterprise Science and Engineering Division and an Associate Professor with the School of Systems and Enterprises, Stevens Institute of Technology. He has conducted funded research for both private industry and government. He has published over 100 refereed manuscripts in technical journals, book chapters, and industry reports. He has presented his research findings nationally and internationally at con-

ferences, such as INFORMS, ISERC, INCOSE, CESUN, and ESREL. He is a member of the Technical Committee on System Reliability for European Safety and Reliability Association and an Associate Editor of IISE Transactions. He has served as the President for the Quality Control and Reliability Division Board of the Institute of Industrial and Systems Engineers.

...